Author's response to reviews

Title: Bile reflux index after therapeutic biliary procedures

Authors:

Sedef Kuran (sedefozdal@hotmail.com)
Erkan Parlak (drerkanparlak@yahoo.com.tr)
Gulden Aydog (Gulden476@hotmail.com)
Sabite Kacar (sabkacar@yahoo.com)
Nurgul Sasmaz (sasmaznur@yahoo.com)
Ali Özden (ankaragastro@yahoo.com)
Burhan Sahin (drburhansahin@yahoo.com)

Version: 4 Date: 2 February 2008

Author's response to reviews:

1. The sentence case was used in the title
   `¿Bile reflux index after therapeutic biliary procedures `¿
2. Section headings and subheadings was converted to sentence case
3. The key words were removed from abstract
4. Authors’ contributions- was corrected as the format (initials were used to refer to each author’s contribution
5. Authors’ contributions `¿`¿All authors read and approved the final manuscript `¿`¿ was added